Author’s response to reviews

Title: Sexual and reproductive health services for women with disability: a qualitative study with service providers in the Philippines

Authors:

Kira Lee (kira.yy.lee@gmail.com)
Alexandra Devine (devinea@unimelb.edu.au)
Ma. Jesusa Marco (ma.jesusa.marco@dlsu.edu.ph)
Jerome Zayas (jeromebzayas@gmail.com)
Liz Gill-Atkinson (liz.gill-atkinson@unimelb.edu.au)
Cathy Vaughan (c.vaughan@unimelb.edu.au)

Version: 2 Date: 15 June 2015

Author’s response to reviews:

Dear April,

I have included the section on no competing interests in the manuscript now.

Best wishes,

Cathy